CHARACTERIZATION OF CARBON NANO PARTICULATE METAL MATRIX COMPOSITES USING EXPERIMENTAL, ANALYTICAL AND FEA METHODS

T. Prasad

Ph.D (Reg. No: 0803PH1567), Department of Mechanical Engineering, JNTUH College of Engineering, Hyderabad



Under the Guidance of Dr. A. Chennakesava Reddy, Professor, JNT University Hyderabad

ABSTRACT

From the past few decades researchers tried numerous combinations of matrices and reinforcements since work strictly on MMCs began in the 1950s. This led to developments for aerospace and defense applications, but resultant commercial applications were limited. In the last 20 years, MMCs evolved from laboratories to a class of materials with numerous applications and commercial markets. After the collapse of the Berlin Wall, prevailing order in the world changed drastically and lead to the development of metal matrix composites.

Composite materials have been recognized over a long period for characterization of AA6061, AA6063 and AA7020 composites in the application of aerospace vehicles, automobile industries due to light weight and high strength. AA6061 is a precipitation hardening aluminum alloy containing magnesium and silicon as its major alloying elements. AA6063 alloy possesses good mechanical properties and enables heat treatment whereas AA7020 alloy is also a heat treatable alloy.

In this work various metal matrix composites are prepared with AA6061, AA6063 and AA7020 as base metals and carbon black nano powder as reinforcement in proportions of 10%, 20% and 30% respectively. In this work two-step stir casting method has been used for manufacturing of Al-alloy Metal Matrix Composites with reinforcement of carbon black nano particles. Before testing the Al alloy-MMC samples, a solution treatment was applied at 5000C for 1 hour, followed by quenching in cold water. The composite

samples were then aged at ambient temperature for 100 hours. The tensile specimens were placed in the grips of a Universal Test Machine (UTM) of 20tonne capacity at a specified grip separation and pulled until failure. Wear test was performed in a pin on disc wear apparatus under different loads of 10, 20, 30N with variation in sliding speeds and sliding distances. The Micro structure analysis of these composites has been studied with the help of Scanning Electron Microscope (SEM). The Effects of carbon black amount on fracture and wear behavior of these AA6061, AA6063 and AA7020/carbon black metal matrix composites produced by stir casting route were investigated. Wear test were performed in a pin on disc wear apparatus under different loads of 10, 20, 30N with different sliding speeds and at different sliding distances. Finally experimental values are validated with finite element analysis software.

The major conclusions of this research work are given below:

1. Apart AA6061/CB, AA6063/CB and AA7020/CB metal matrix composites with different volume fraction of carbon black nanoparticles, AA 7070/CB metal matrix composites with 30% volume fraction of carbon black nano particulates found to have greatest tensile strength. the load transfer from the matrix material to CB nano particulates was 45 MPa.

2. The wear studies on the metal matrix composites showed a decreasing trend of wear rate with respect to increase in volume fraction of carbon black nano particulates. The maximum sliding wear of AA6061/CB, AA6063/CB and AA7020/CB metal matrix composites were respectively, 68.72%, 65.20% and 60.07% and attributed to the sliding distance.

1. Scope of the Research Work

There has been a continuous effort by researchers to improve the properties of materials to meet the modern technological requirements and demands of consumers for the development of advanced material systems which are more energy-efficient, lighter by weight, stronger and economical (Davis, 1993; Foltz, 2001; Anthony, 2012). The development of composite materials is one

of such efforts that have changed the world of materials. The composite

comprises two or more constituent materials that are microscopically combined but not soluble in each other. The materials used for the

reinforcement are in the form of fibers, flakes, or particulates. The materials used for the matrix are by and large continuous (Chawla, 2001; Surappa, 1981).

During the last decade, many researchers made investigations on the metalmatrix composites (MMCs) with the use of various metallic materials. The important metals which are found to be useful for the application to industries

are Al, Mg, Ti, Cu, and their alloys. Though there are various types of MMCs,

aluminium and its alloys as matrix material have been extensively employed in most of the recent applications in aerospace, automobiles, and many other engineering and domestic fields. This has been motivated by a need to decrease energy consumption in socio-economical and industrial practice with their excellent strength and low weight features (Hashim, 2002; Bayraktar, 2010). There are significant variances in published data on the properties of MMCs. This is attributed to the reality that there are no industry

standards for MMCs.

Different types of reinforcements including continuous fibers, discontinuous fibers, whiskers, and particulates, from 10 to 60% vol. have been investigated by Foltz and Blackmon (1990); Reddy and Essa Zitoun (2009, 2010), and many others. The choice of reinforcement material is very important to manufacture MMCs (Mazaherin, 2013). The reinforcement material should be such that it not only should improve the properties but also should reduce the manufacturing cost of the composites. This thesis would thus explore the

choice of cheaper reinforcement material, namely carbon black (CB) as shown in figure 1 to produce Al alloy-MMCs while achieving the required mechanical properties.



Figure 1: Carbon black

2. Research Justification

This research was aimed at the manufacture of Al alloy-MMCs using carbon

black (CB) as reinforcement particulate resulted from the incomplete combustion of petroleum products. This research was also concerned to validate CB reinforced MMCs for wear-resistant applications. In general, particulate reinforced Al-MMCs (PR-Al-MMCs) are manufactured by the stir casting method. In this method, accomplishing a uniform distribution of reinforcement particulates within the matrix alloy is a principal task that influences directly the properties and the quality of Al alloy-MMCs as shown in figure 2. In the case of (PR-Al-MMCs), the dispersal of the reinforced particulates in the matrix alloy is affected by numerous factors during stir casting. These consist of the rheological performance of the liquid matrix alloy, the reinforcement mixing method, chemical reactions of the reinforcement and the matrix, and altering reinforcement distribution through solidification.



Figure 2: Uniform and non-uniform distribution of particulates in MMCs.

The constituents namely, matrix and reinforcement of the MMC should bond strongly with each other without any adverse chemical reactions. In Al-MMCs reinforced with non-metal carbides such as silicon carbide, boron

carbide, etc., or carbon fibers, the aluminium carbide (Al_4C_3) often forms as an unwanted product or compound as per studies conceded by Greenwood and Earnshaw (1997). The chemical reaction of carbon and aluminium is

shown in figure 3. Aluminium carbide (Al₄C₃) particles which are finely spread

in the Al-alloy matrix, diminishes the susceptibility of the material to creep as stated by Zhu et al. (1998).

$$4A1 + 3C \xrightarrow{f^{\circ}C} Al_4C_3$$

(1.1)



Figure 3: Product formed during the reaction between carbon and aluminium.

Ph.D Thesis Department of Mechanical Engineering, JNTUH College of Engineering, JNT University, Hyderabad



Figure 4: Effect of particulate size on the elastic modulus of the composite.

Cocen and Onel (2002) have reported that the microstructures of as-cast Al-SiC MMCs, manufactured by the stir casting method, show a uniform distribution of SiC particulates with some local clusters and some degree of porosity. During the microstructural investigation of SiC particulates reinforced AA6063 alloy-MMCs, Alaneme and Aluko (2012) have detected that the volume per cent of SiC particulates would not stimulus its pattern of distribution in the as-cast condition. Also, the elastic modulus of PR-Al-MMCs is insensitive to particle size as stated by Jing et. al (2002). However, when the particulate size is reduced to a critical size of 20 nm, there would be the effect of particulate size on the elastic modulus as shown in figure 4.

Boopathi et al. (2013) have observed non-uniformity in the distribution of reinforced particulates in the case of Al-SiC, Al-fly ash, and Al-SiC-fly ash composites produced by stir casting method. The wettability at the interface between the matrix and reinforced particulates plays another important role in determining the mechanical properties of Al-MMCs. Wettability of the particulates can be boosted by coating the particles with metals such as Ni and Cu, or addition of active elements such as Mg into liquid aluminium, or preheating of the particulates before addition into liquid aluminium (Kolsgaard, 1993). In the present research work, nickel-coated carbon black nanoparticulates were used.

The durability, emissions, and fuel economy of internal combustion engines are associated with lubrication efficiency. Ineffective lubrication can lead to friction and wear losses, which could negatively impact the fuel consumption and the lifetime of engines. Solid lubricants lodge an exceptional place in enhancing wear-resistance in situations where the liquid lubricants are inept like in aerospace or automotive applications. Normally, the size of reinforced particulates influences mechanical properties such as strength, stiffness, ductility, and fracture of self-lubricating composites. MMCs reinforced by larger particulates are vulnerable to the formation of defects, such as porosity and cracking, which results in premature failure of the composites. But, it is expected to have excellent properties when the reinforced particulate size is in the nano-scale (Bakunin, et. al, 2004; Tevet, et. al, 2011).

To reach the said goals of the present research work, the foremost difficulties such as the uniform distribution of the reinforced particulates in the Al-alloy,

wettability with the Al-alloy, and the reaction of carbon black (CB) with the Al-alloy are to be addressed. The research work was also aimed at the self-

lubricating capacity of nano-scaled CB reinforced Al alloy-based MMCs. The experimental results were also validated using the finite element approach.

3. Aim and Objectives

This research aims to develop Al alloy-based MMCs reinforced with carbon

black nanoparticulates to understand the material properties, process characteristics, microstructural characteristics needed to obtain optimal and reliable properties. The key objectives include:

(i) Development of composite materials from AA6061, AA6063, and AA7020 alloys reinforced with the carbon black (CB) nanoparticulates by using stir casting methods.

(ii) Evaluation of the effect of the CB nanoparticulates on the mechanical and

metallurgical properties.

(iii) Evaluation of the effect of the CB nanoparticulates on the tribological properties.

proportion.

(iv) Understanding the reaction of the CB nanoparticulates with the Al alloys.

(v) Validation of experiment results with those obtained from the finite element analysis (FEA).

4. Overview of the Thesis

The thesis is organized into the following chapters mentioned below:

Chapter 1 is an outline of the thesis.

Chapter 2 is to review and critical examination of relevant literature to explore the use of carbon black as the reinforcement particulate.

Chapter 3 focuses the experimentation and methodology adopted for the research.

Chapter 4 discusses the results of mechanical and metallurgical experimentation. It also includes the validation of experimental test results

with FEA results.

Chapter 5 discusses the results of tribological experimentation.

Chapter 6 focuses on the conclusions and introduces areas for future research.

5. Summary

The composite material comprises two or more physically and chemically distinct constituents namely, matrix and reinforcement. The MMC is an

engineered blend of metal (matrix) and hard particulate reinforcement to get

explicit properties such as tensile strength, wear resistance, corrosion resistance, etc., for various engineering applications. In Al-MMCs the

reinforced non-metal carbides (silicon carbide, boron carbide, etc.) or carbon

fibers, aluminium carbide every so often form precipitated compounds. The

vivacious problems during the manufacturing of PR-Al-MMCs are uniform distribution and wettability of reinforced particulates. Also, more resourceful lubrication requires accomplishing the highly demanding working conditions of automotive engines. It aimed the present work at manufacture and explore the performance of PR-Al-MMCs based on aluminium alloy as a matrix and carbon black (CB) nanoparticulates as a reinforcement substitute for carbides. It also deliberated the present research wok to validate experimental

results with those found by empirical models and finite element analysis (FEA).

References

- 1. A. Chennakesava Reddy, Temperature and Anisotropy Induced Micromechanics for Negative Poisson's Ratio h-BN/5050 Al Alloy Composites, International Journal of Engineering and Technology, Vol.9, No.4, pp.2846-2853, 2017.
- 2. A. Chennakesava Reddy, Low and High Temperature Micromechanical Behavior of BN/3003 Aluminum Alloy Nanocomposites, International Journal of Mechanical Engineering and Technology, Vol.6, No.4, pp.27-34, 2017.
- 3. A. Chennakesava Reddy, Consequences of Magnesium in 5050 Aluminum Alloy on Wettability, Strengthening Mechanisms and Fracture Behavior of Silicon Carbide Nanoparticle Metal Matrix Composites, International Journal of Research in Mechanical engineering & Technology, Vol.7, No.1, pp.89-96, 2017.
- Chennakesava R Alavala, Thermal Expansion Behavior of Al/Magnesia Metal Matrix Composites, International Journal of Science and Research, Vol.5, No.8, pp.1817-1821, 2016.

- 5. Chennakesava R Alavala, Nano-mechanical modeling of thermoelastic behavior of AA6061/silicon oxide nanoparticulate metal matrix composites, International Journal of Science and Research, Vol.5, No.1, pp.550-553, 2016.
- 6. Seelam Pitchi Reddy, A. Chennakesava Reddy, Tensile and Flexural Strength of OKRA Fiber Reinforced Polymer Composites, International Journal of Engineering and Management Research, Vol.6, No.1, pp.491-495, 2016.
- K. Shiva Kumar, A. Chennakesava Reddy, Study on Reinforcement Materials for Nylon Matrix Composites - A Review, International Journal of Scientific & Engineering Research, Vol.7, No.6, pp.156-160, 2016.
- 8. A. Chennakesava Reddy, Influence of Stiffeners on Strength of E-Glass/Epoxy Composite Submergible Hull Subjected to Shock Pressure Load using Finite Element Method, International Conference on Advancements in Materials for Manufacturing, Hyderabad, 2016.
- 9. Chennakesava R Alavala, Nanomodeling of nonlinear thermoelastic behavior of AA5454/ silicon nitride nanoparticulate metal matrix composites, International Journal of Engineering Research and Application, Vol.6, No.1, pp.104-109, 2016.
- Chennakesava R Alavala, Thermoelastic Behavior of Nanoparticulate BN/AA5050 Alloy Metal Matrix Composites, International Journal of Engineering and Advanced Research Technology, Vol.2, No.1, pp.6-8, 2016.
- 11. Chennakesava R Alavala, Micromechanical Modelling of Thermoelastic Behavior of AA7020/TiC Metal Matrix Composites, International Journal of Scientific Engineering and Research, Vol.4, No.2, pp.1-5, 2016.
- 12. T. Prasad, A. Chennakesava Reddy, Micro-Tensile Behavior of AA7020/Carbon Black Nanoparticle Metal Matrix Composites, International Journal of Engineering and Science, Vol.6, No.8, pp.36-40, 2016.
- 13. Chennakesava R Alavala, Micromechanics of Thermoelastic Behavior of AA2024/MgO Metal Matrix Composites, International Journal of Advanced Technology in Engineering and Science, Vol.4, No.1, pp.33-40, 2016.
- 14. Chennakesava R Alavala, Comparison of Experimental and Theoretical CTE of Al/h-BN Metal Matrix Composites, International Journal of Material Sciences and Technology, Vol.6, No.1, pp.13-20, 2016.
- Chennakesava R Alavala, Micromechanics of Thermoelastic Behavior of AA6070 Alloy/Zirconium Oxide Nanoparticle Metal Matrix Composites, International Journal of Engineering Research & Science, Vol.2, No.2, pp.1-8, 2016.
- A. Chennakesava Reddy, Effect of Yttrium Oxide Doping on CTE of Al/ZrO2 Metal Matrix Composites, IOSR Journal of Mechanical and Civil Engineering, Vol.13, No.5, pp.93-98, 2016.
- P. Pavani, A. Chennakesava Reddy, Micromechanics of Thermoelastic Behavior of Polypropylene/Calcite (Modified with Triton X-100) Nanocomposites, International Journal of Science and Research, Vol.5, No.2, pp.1003-1006, 2016.
- 18. Chennakesava R Alavala, Effect of Thermoelastic Behavior on interfacial debonding and Particulate Fracture in AA1100/TiN Nanoparticulate Metal Matrix Composites, International Journal of Science and Research, Vol.5, No.3, pp.1295-1300, 2016.

- 19. Chennakesava R Alavala, Influence of Temperature on Particulate Fracture of AA2024 Alloy/Titanium Oxide nanoparticulate Metal Matrix Composites, International Journal of Scientific Engineering and Applied Science, Vol.2, No.4, pp.1-6, 2016.
- Chennakesava R Alavala, Influence of CTE Mismatch on Debonding and Particulate Damage in AA1100 Alloy/ZrC Nanoparticulate Metal Matrix Composites, International Journal of Innovative Research in Science, Engineering and Technology, Vol.5, No.3, pp.3489-3495, 2016.
- 21. A. Chennakesava Reddy, Design and Finite Element Analysis of E-glass Fiber Reinforced Epoxy Composite Air Bottle used in Missile System: Experimental Validation, International Journal of Scientific & Engineering Research, Vol.6, No.8, pp.157-165, 2015.
- 22. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics between Matrix and Reinforced Nanoparticle of AA2124/AlN Nanocomposites: Mathematical and Experimental Validation, International Journal of Engineering and Advanced Technology, Vol.5, No.1, pp.5-12, 2015.
- 23. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics Between Matrix and Reinforced Nanoparticle of AA3105/AlN Nanocomposites, International Journal of Mechanical Engineering, Vol.4, No.5, pp.25-36, 2015.
- 24. A. Chennakesava Reddy, Shock Analysis of E-Glass/Epoxy Composite Submersible Hull Subjected to Pressure Loads of Underwater Explosion using Finite Element Method -Experimental Validation, International Journal of Scientific & Engineering Research, Vol.6, No.9, pp.1461-1468, 2015.
- 25. A. Chennakesava Reddy, Consequences of Interphase between Matrix and Reinforced Nanoparticle on Behavior of AA6262/AlN Nanocomposites, International Journal of Science and Research, Vol.4, No.9, pp.1045-1049, 2015.
- 26. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics Between Matrix and Reinforced Nanoparticle of AA4032/AlN Nanocomposites, International journal of research in mechanical engineering, Vol.3, No.5, pp.13-21, 2015.
- 27. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics between Matrix and Reinforced Nanoparticle of AA5154/AlN Nanocomposites, International Journal of Advanced Research, Vol.3, No.9, pp.703-710, 2015.
- 28. A. Chennakesava Reddy, Reduction of Vibrations and Noise using Nylon-66/Al2O3 Nanocomposite Gear Box in Lathe, International Journal of Science and Research, Vol.4, No.9, pp.1476-1480, 2015.
- 29. A. Chennakesava Reddy, Reduction of Vibrations and Noise using Nylon-66/SiC Nanocomposite Gear Box in Lathe, International Journal of Science and Research, Vol.4, No.9, pp.1485-1489, 2015.
- 30. A. Chennakesava Reddy, Reduction of Vibrations and Noise using Nylon-66/Fe2O3 Nanocomposite Gear Box in Lathe, International Journal of Science and Research, Vol.4, No.9, pp.1490-1494, 2015.
- 31. A. Chennakesava Reddy, Reduction of Vibrations and Noise using AA7020/Al2O3 Nanocomposite Gear Box in Lathe, International Journal of Scientific & Engineering Research, Vol.6, No.9, pp.671-677, 2015.

- 32. A. Chennakesava Reddy, Reduction of Vibrations and Noise using AA7020/SiC Nanocomposite Gear Box in Lathe, International Journal of Scientific & Engineering Research, Vol.6, No.9, pp.678-684, 2015.
- 33. A. Chennakesava Reddy, Reduction of Vibrations and Noise using AA7020/Fe2O3 Nanocomposite Gear Box in Lathe, International Journal of Scientific & Engineering Research, Vol.6, No.9, pp.685-691, 2015.
- A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics between Matrix and Reinforced Nanoparticle of AA8090/AlN Nanocomposites, Asian Journal of Engineering and Technology, Vol.3, No.5, pp.505-511, 2015.
- 35. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics between Matrix and Reinforced Nanoparticle of AA6061/AlN Nanocomposites, International Journal of Nanotechnology and Application, Vol.5, No.5, pp.1-10, 2015.
- 36. A. Chennakesava Reddy, Estimation of Thermoelastic Behavior of Three-phase: AA1100/Ni-Coated Boron Carbide Nanoparticle Metal Matrix Composites, International Journal of Scientific & Engineering Research, Vol.6, No.10, pp.662-667, 2015.
- 37. A. Chennakesava Reddy, Effects of Adhesive and Interphase Characteristics between Matrix and Reinforced Nanoparticle of AA7175/AlN Nanocomposites, International Journal of Scientific Engineering and Research, Vol.3, No.11, pp.95-98, 2015.
- 38. A. Chennakesava Reddy, Influence of volume fraction, size, cracking, clustering of particulates and porosity on the strength and stiffness of 6063/SiCp metal matrix composites, International Journal of Research in Engineering and Technology, Vol.4, No.1, pp.1-9, 2015.
- 39. A. Chennakesava Reddy, Studies on loading, cracking and clustering of particulates on the strength and stiffness of 7020/SiCp metal matrix composites, International Journal of Metallurgical & Materials Science and Engineering, Vol.5, No.1, pp.53-65, 2015.
- 40. A. Chenna kesava Reddy, Cause and Catastrophe of Strengthening Mechanisms in 6061/Al2O3 Composites Prepared by Stir Casting Process and Validation Using FEA, International Journal of Science and Research, Vol.4, No.2, pp.1272-1281, 2015.
- 41. A. Chenna kesava Reddy, Influence of Particle Size, Precipitates, Particle Cracking, Porosity and Clustering of Particles on Tensile Strength of 6061/SiCp Metal Matrix Composites and Validation Using FEA, International Journal of Material Sciences and Manufacturing Engineering, Vol.42, No.1, pp.1176-1186, 2015.
- 42. A. Chennakesava Reddy, Evaluation of Curing Process for Carbon-Epoxy Composites by Mechanical Characterization for Re-entry Vechicle Structure, International Journal of Scientific & Engineering Research, Vol.6, No.3, pp.65-70, 2015.
- 43. A.Chennakesava Reddy, B. Kotiveerachari, P. Rami Reddy, Saving of Thermal Energy in Air-Gap Insulated Pistons Using Different Composite Materials for Crowns, International Journal of Scientific & Engineering Research, Vol.6, No.3, pp.71-74, 2015.
- 44. A. Chennakesava Reddy, Cause and Catastrophe of Strengthening Mechanisms in 6063/Al2O3 Composites Prepared by Stir Casting Process: Validation through FEA, International Journal of Scientific & Engineering Research, Vol.6, No.3, pp.75-83, 2015.
- 45. A. Chenaakesava Reddy, Cause and catastrophe of strengthening mechanisms in 7020/Al2O3 composites prepared by stir casting process and validation through FEA, International Journal of Advanced Research, Vol.3, No.3, pp.603-614, 2015.

- 46. A. Chennakesava Reddy, Evaluation of Curing Process for Kevlar 49-Epoxy Composites by Mechanical Characterization Designed for Brake Liners, International Journal of Science and Research, Vol.4, No.4, pp.2365-2371, 2015.
- 47. A. Chennakesava Reddy, Evaluation of Curing Process for Bi-directional S-Glass (5HS)/Epoxy (780E +782H) Composites Fabricated by Vacuum Infusion Process for Wind Energy Blades, International Journal of Advanced Research, Vol.3, No.4, pp.667-675, 2015.
- 48. Karan Agarwal, Nirmala Akhi, Regalla Srinivas, A. Chennakesava Reddy, Enhancement in Mechanical Behavior of Nylon/Teflon Composites by Addition of Nano Iron Oxide (Fe2O3), International Journal of Science and Research, Vol.4, No.5, pp.927-932, 2015.
- 49. A. Chennakesava Reddy, Characterization of Mechanical Behavior of Nylon/Teflon Nano Particulate Composites, International Journal of Advanced Research, Vol.3, No.5, pp.1241-1246, 2015.
- 50. A. Chennakesava Reddy, Necessity of Strain Hardening to Augment Load Bearing Capacity of AA1050/AIN Nanocomposites, International Journal of Advanced Research, Vol.3, No.6, pp.1211-1219, 2015.
- 51. T. Prasad, A. Chennakesava Reddy, Effects of Adhesive Characteristics between Matrix and Reinforced Nanoparticle of AA6061/Carbon Black Nanocomposites, International Journal of Scientific & Engineering Research, Vol.6, No.7, pp.40-45, 2015.
- 52. T. Prasad, A. Chennakesava Reddy, Effects of Adhesive Characteristics between Matrix and Reinforced Nanoparticle of AA6063/Carbon Black Nanocomposites, International Journal of Science and Research, Vol.4, No.7, pp.1777-1781, 2015.
- 53. A. Chennakesava Reddy, Influence of Interphase on Tensile Behavior of Strain Hardened AA1100/AlN Nanocomposites Using RVE Models and Experimental Validation, International Journal of Engineering, Science and Technology, Vol.7, No.7, pp.239-250, 2015.
- 54. T. Prasad, A. Chennakesava Reddy, S. Jushkumar, Tensile and fracture behavior of 6061 Al-SiCp metal matrix composites, International Conference on Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.38-44, Paramount Publishing House, 2014.
- 55. Essa Zitoun, A. Chennakesava Reddy, Metallurgical characteristics of fracture behaviour in Al/SiC metal matrix composite, International Conference on Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.59-66, Paramount Publishing House, 2014.
- 56. G. Satish Babu, A. Chennakesava Reddy, Fracture behavior of alumina particles reinforced with different matrix aluminium alloys, International Conference on Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.67-74, Paramount Publishing House, 2014.
- 57. S. Pichi Reddy, P. V. Chandra Sekhar Rao, A. Chennakesava Reddy, G. Parmeswari, Tensile and flexural strength of glass fiber epoxy composites, International Conference on Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.91-95, Paramount Publishing House, 2014.
- 58. S.Sreenivasulu, A. chennakeshava Reddy, Thermo-mechanical properties of silicon nitrate ceramic composites for fused deposition modeling, International Conference on

Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.153-166, Paramount Publishing House, 2014.

- 59. P. Laxminarayana, A. Chennakesava Reddy, Influence of heat treatment on mechanical behavior of aluminium- 7075/silcon carbide composites manufactured by squeeze casting process, International Conference on Advanced Materials and Manufacturing Technologies, JNTUH Hyderabad, 9789382163466, pp.167-177, Paramount Publishing House, 2014.
- S. Sreenivasulu, A. Chennakesava Reddy, Mechanical Properties Evaluation of Bamboo Fiber Reinforced Composite, International Journal of Engineering Research, Vol.3, No.1, pp.187-194, 2014.
- 61. T. Prasad, A. Chennakesava Reddy, T.Tirupati, Material Characterization Of 6061 Al-SiCp Metal Matrix Composites, International Journal of Mathematical Sciences, Technology and Humanities, Vol.3, No.1, pp.756-765, 2013.
- 62. A. Chennakesava Reddy, Effect of Phase Transformation from h-BN to c-BN on Nanoparticle Fracture Tendency in AA8090/h-BN Particle-Reinforced Metal Matrix Composites, 5th International Conference on Modern Materials and Manufacturing, Bangalore, pp.365-372, 2013.
- 63. A. Chennakesava Reddy, Combined Loading and Micromechanical Analysis of AA5050 Alloy-Silicon Oxide Particle-Reinforced Metal Matrix Composites, 5th International Conference on Modern Materials and Manufacturing, Bangalore, pp.373-378, 2013.
- 64. A. Chennakesava Reddy, Thermal Expansion of Al Matrix Composites Reinforced with TiN Nanoparticles, 2nd International Conference on Thermal and Tribological Behavior of Composites, New Delhi, pp.144-148, 2013.
- 65. T. Prasad, A. Chennakesava Reddy, S. Madhava Reddy, N. Arjun, Experimental investigation of mechanical behaviour of glass-epoxy composites, International Conference on Recent Advances in Material Processing Technology, Kovilpatti, 2013.
- 66. S. Pichi Reddy, B. Ramana, A. Chennakesava Reddy, Sintering Characteristics of Al-Pb/Fly-Ash Metal Matrix Composites, Transactions Indian Institute Metals, 0972-2815, Vol No.66, Issue No.1, pp.87-95, Indian Institute of Metals, 2013.
- 67. M. S. Ramgir, A. Chennakesava Reddy, Effect of Thermal-heating on Nanoparticle Fracture Trend in AA2024/c-BN Particle-Reinforced Metal Matrix Composites, 4th International Conference on Modern Materials and Manufacturing, Chennai, pp.305-308, 2012.
- M. S. Ramgir, A. Chennakesava Reddy, Effect of Thermo-Tensile Loading on Micromechanical Behavior of AA6061 Alloy-Titanium Carbide Composites, 4th International Conference on Modern Materials and Manufacturing, Chennai, pp.309-313, 2012.
- 69. S. Pitchi Reddy, B.Ramana, A. Chennakesava Reddy, Sintered Density and Porosity of Al-15wt% Pb/Fly-ash Metal Matrix Composites, International Journal of Engineering and Materials Sciences, Vol.5, No.1, pp.59-66, 2012.
- 70. A. S. Goud, A. Chennakesava Reddy, Evaluation of Nanoparticle Fracture in MgO Reinforced Aluminum matrix composites, 3rd International Conference on Modern Materials and Manufacturing, New Delhi, pp.320-324, 2011.

- 71. A. S. Goud, A. Chennakesava Reddy, Interface Failure Analysis of TiB2 Reinforced Aluminum Alloy Matrix Composites, 3rd International Conference on Modern Materials and Manufacturing, New Delhi, pp.325-328, 2011.
- 72. A. Chennakesava Reddy, Essa Zitoun, Tensile properties and fracture behavior of 6061/Al2O3 metal matrix composites fabricated by low pressure die casting process, International Journal of Materials Sciences, Vol.06, No.02, pp.147-157, 2011.
- 73. A. Chennakesava Reddy, Influence of strain rate and temperature on superplastic behavior of sinter forged Al6061/SiC metal matrix composites, International Journal of Engineering Research & Technology, Vol.4, No.2, pp.1189-198, 2011.
- 74. A. Chennakesava Reddy, Essa Zitoun, Strengthening mechanisms and fracture behavior of 7072Al/Al2O3 metal matrix composites, International Journal of Engineering Science and Technology, Vol.3, No.7, pp.6090-6100, 2011.
- 75. S. Pitchi Reddy, B.Ramana, A. Chennakesava Reddy, Determination green hardness and strength al-15% Pb-fly ash metal matrix composites, International Journal of Emerging Technologies and Applications in Engineering Technology and Sciences, Vol.4, No.2, pp.11-14, 2011.
- 76. A. Chennakesava Reddy, Evaluation of mechanical behavior of Al-alloy/Al2O3 metal matrix composites with respect to their constituents using Taguchi, International Journal of Emerging Technologies and Applications in Engineering Technology and Sciences, Vol.4, No.2, pp.26-30, 2011.
- 77. M. S. Ramgir, A. Chennakesava Reddy, Control of B4C Reinforced Particulates on Dry Wear Resistance of AA2024/B4C Composites, 3rd International Conference on Modern Materials and Manufacturing, New Delhi, pp.336-340, 2011.
- 78. A. Chennakesava Reddy, Tensile fracture behavior of 7072/SiCp metal matrix composites fabricated by gravity die casting process, Materials Technology, Vol.26, No.5, pp.257-262, 2011.
- 79. A. Chennakesava Reddy, Evaluation of mechanical behavior of Al-alloy/SiC metal matrix composites with respect to their constituents using Taguchi techniques, i-manager's Journal of Mechanical Engineering, Vol.1, No.2, pp.31-41, 2011.
- 80. A. Chennakesava Reddy, B. Kotiveerachari, Influence of microstructural changes caused by ageing on wear behaviour of Al6061/SiC composites, Journal of Metallurgy & Materials Science, Vol.53, No.1, pp.31-39, 2011.
- M. Mastanaiah, A. Chennakesava Reddy, Abrasive Wear of AA3003/ZrC Composites, 3rd International Conference on Modern Materials and Manufacturing, New Delhi, pp.347-351, 2011.
- V. K. Reddy, A. Chennakesava Reddy, Unlubricated Sliding of AA4015/TiB2 Metal Matrix Composites, 3rd International Conference on Modern Materials and Manufacturing, New Delhi, pp.352-356, 2011.
- 83. Y. S. A. Kumar, A. Chennakesava Reddy, Fabrication and Properties of AA7020-TiN Composites under Combined Loading of Temperature and Tension, 2nd International Conference on Modern Materials and Manufacturing, Pune, pp.276-280, 2010.
- 84. G. V. R. Kumar, A. Chennakesava Reddy, Wear Equations for AA7020-MgO Composites Sliding Against En32 Steel Disc, 2nd International Conference on Modern Materials and Manufacturing, Pune, pp.294-300, 2010.

- 85. S. Pitchi Reddy, B.Ramana, A. Chennakesava Reddy, Compacting Characteristics of Aluminum 10 wt% Fly Ash Lead Metal Matrix Composites, International Journal of Materials Science, Vol.5, No.6, pp.777-783, 2010.
- 86. A. Chennakesava Reddy, B. Kotiveerachari, Effect of aging condition on structure and the properties of Al-alloy / SiC composite, International Journal of Engineering and Technology, Vol.2, No.6, pp.462-465, 2010.
- 87. A. Chennakesava Reddy, Tensile properties and fracture behavior of 6063/SiCP metal matrix composites fabricated by investment casting process, International Journal of Mechanical Engineering and Materials Sciences, Vol.3, No.1, pp.73-78, 2010.
- 88. A. Chennakesava Reddy, M. Vidya Sagar, Two-dimensional theoretical modeling of anisotropic wear in carbon/epoxy FRP composites: comparison with experimental data, International Journal of Theoretical and Applied Mechanics, Vol.6, No.1, pp.47-57, 2010.
- 89. A. Chennakesava Reddy, Essa Zitoun, Tensile behavior of 6063/Al2O3 particulate metal matrix composites fabricated by investment casting process, International Journal of Applied Engineering Research, Vol.1, No.2, pp.542-552, 2010.
- S. Pitchi Reddy, B. Ramana, A. Chennakesava Reddy, Compacting Characteristics of Al-15%Pb - Flyash Metal Matrix Composites, Journal of Manufacturing Engineering, Vol.5, No.1, pp.55-59, 2010.
- 91. A. Chennakesava Reddy, Essa Zitoun, Matrix Al-alloys for silicon carbide particle reinforced metal matrix composites, Indian Journal of Science and Technology, Vol.3, No.12, pp.1184-1187, 2010.
- 92. A. Chennakesava Reddy, Mechanical properties and fracture behavior of 6061/SiCp Metal Matrix Composites Fabricated by Low Pressure Die Casting Process, Journal of Manufacturing Technology Research, Vol.1, No.3&4, pp.273-286, 2009.
- 93. A. Chennakesava Reddy, Essa Zitoun, Matrix Al-alloys for alumina particle reinforced metal matrix composites, Indian Foundry Journal, Vol.55, No.1, pp.12-16, 2009.
- 94. S. Pitchi Reddy, A. Chennakesava Reddy, Synthesis and Characterization of Zirconium Carbide Nanoparticles Reinforced AA2024 Alloy Matrix Composites Cast by Bottom-Up Pouring, 7th International Conference on Composite Materials and Characterization, Bangalore, pp.211-215, 2009.
- 95. Essa Zitoun, A. Chennakesava Reddy, Analysis of Micromechanical Behavior of AA3003 Alloy - Graphite Metal Matrix Composites Cast by Bottom-Up Pouring with Regard to Agglomeration and Porosity, 7th International Conference on Composite Materials and Characterization, Bangalore, pp.216-220, 2009.
- 96. P. Rami Reddy, A. Chennakesava Reddy, Processing of AA4015-Zirconium Oxide Particulate Metal Matrix Composites by Stir Casting Technology, 7th International Conference on Composite Materials and Characterization, Bangalore, pp.221-224, 2009.
- 97. B. Kotiveera Chari, A. Chennakesava Reddy, Bottom-Up Pouring and its Effect on Porosity and Clustering in Casting of AA1100/Silicon Nitride Particle-Reinforced Metal Matrix Composites, 6th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.110-114, 2008.
- 98. Essa Zitoun, A. Chennakesava Reddy, Microstructure-Property Relationship of AA3003/Boron Nitride Particle-Reinforced Metal Matrix Composites Cast by Bottom-Up

Pouring, 6th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.115-119, 2008.

- 99. A. Chennakesava Reddy, Wear and Mechanical Behavior of Bottom-Up Poured AA4015/Graphite Particle-Reinforced Metal Matrix Composites, 6th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.120-126, 2008.
- 100. S. Pitchi Reddy, A. Chennakesava Reddy, Effect of Needle-like Brittle Intermetallic Phases on Fracture Behavior of Bottom-up Poured AA5050/Titanium Carbide Particle-Reinforced Metal Matrix Composites, 6th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.127-132, 2008.
- A. Chennakesava Reddy, Strength and fracture mechanisms in carbon-carbon composites, International symposium on Advanced Materials and Processing, Bagalkot, pp.138-144, 2007.
- 102. A. Chennakesava Reddy, Role of Porosity and Clustering on Performance of AA1100/Boron Carbide Particle-Reinforced Metal Matrix Composites, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.122-127, 2007.
- 103. P. Rami Reddy, A. Chennakesava Reddy, Formation of Gas Porosity and Clustering in Stir Cast AA2024/Titanium Diboride Particle-Reinforced Metal Matrix Composites and Influence on Micromechanical Properties, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.128-132, 2007.
- 104. P. Rami Reddy, A. Chennakesava Reddy, Structure and Properties of Liquid Metal Processed Zirconium Oxide Reinforced AA3003 Alloy, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.133-138, 2007.
- 105. A. C. S. Kumar, A. Chennakesava Reddy, Processing of AA4015-Silicon Oxide Particulate Metal Matrix Composites by Stir Casting Technology, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.139-143, 2007.
- 106. A. C. S. Kumar, A. Chennakesava Reddy, Microstructural and Numerical Evaluation of Porosity and Clustering Control over Micromechanical Properties of Cast Titanium Nitride Reinforced AA5050 Alloy, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.144-148, 2007.
- 107. A. Chennakesava Reddy, Effect of Clustering Induced Porosity on Micromechanical Properties of AA6061/Titanium Oxide Particulate Metal matrix Composites, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.149-154, 2007.
- 108. S. Satyanarayana, A. Chennakesava Reddy, High pressure Die Casting of AA7020/Zirconium Carbide Particulate Metal matrix Composites, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.155-159, 2007.
- 109. S. Satyanarayana, A. Chennakesava Reddy, Occurrence of Agglomeration and Porosity during High pressure Die Casting of AA8090/Graphite Particulate Metal matrix Composites, 6th International Conference on Composite Materials and Characterization, Hyderabad, pp.160-164, 2007.
- 110. A. Chennakesava Reddy, Effect of TiC Nanoparticles on the Coefficient of Thermal Expansion Behavior of the Aluminum Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.164-168, 2006.

- 111. S. Madhav Reddy, A. Chennakesava Reddy, Effects of Porosity on Mechanical Properties of Zirconium Oxide/AA1100 Alloy Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.124-128, 2006.
- 112. Essa Zitoun, A. Chennakesava Reddy, High Pressure Die Casting Process on Micromechanical Properties of AA2024/Boron Carbide Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.129-133, 2006.
- 113. Essa Zitoun, A. Chennakesava Reddy, Micromechanical and Porosity Studies of Cast AA3003/ Boron Nitride Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.134-138, 2006.
- 114. A. Chennakesava Reddy, Effect of Porosity Formation during Synthesis of Cast AA4015/Titanium Nitride Particle-Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.139-143, 2006.
- 115. A. Chennakesava Reddy, Stir Casting Process on Porosity Development and Micromechanical Properties of AA5050/Titanium Oxide Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.144-148, 2006.
- 116. A.C. S. Kumar, A. Chennakesava Reddy, Effect of Cold Rolling on Porosity and Micromechanical Properties of AA6061/Zirconium Carbide Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.149-153, 2006.
- 117. S. Madhav Reddy, A. Chennakesava Reddy, Effect of Reinforcement Loading on Porosity and Micromechanical Properties of AA7020/Graphite Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.154-158, 2006.
- 118. A.C. S. Kumar, A. Chennakesava Reddy, Microstructure and Properties of Liquid Metal Processed MgO Reinforced AA8090 Metal Matrix Composites, 5th National Conference on Materials and Manufacturing Processes, Hyderabad, pp.159-163, 2006.
- 119. K. Swapna Sudha, A. C. Reddy, Tensile performance of heat treated AA2024/Al2O3 metal matrix composites using RVE modles: experimental validation, National Conference on Advances in Design Approaches and Production Technologies (ADAPT-2005), pp.332-334, JNTU College of Engineering, Hyderabad, 2005.
- 120. V. K. Prasad, A. C. Reddy, Tensile behavior of tempered AA5050/Al2O3 metal matrix composites using RVE models: experimental validation, National Conference on Advances in Design Approaches and Production Technologies (ADAPT-2005), pp.335-337, JNTU College of Engineering, Hyderabad, 2005.
- 121. M. Chamundeswari, A. C. Reddy, Evaluation of strength improvement in tempered AA5050/SiC metal matrix composites using finite element analysis: experimental validation, National Conference on Advances in Design Approaches and Production Technologies (ADAPT-2005), pp.338-340, JNTU College of Engineering, Hyderabad, 2005.
- 122. S. Sujatha, A. C. Reddy, Assessment of strength improvement in heat treated AA2024/SiC metal matrix composites using finite element analysis: experimental validation, National Conference on Advances in Design Approaches and Production

Technologies (ADAPT-2005), pp.341-343, JNTU College of Engineering, Hyderabad, 2005.

- 123. B. Ramana, A. Chennakesava Reddy, S. Somi Reddy, Fracture analysis of Mg-alloy metal matrix composites, National Conference on Computer Applications in mechanical Engineering, Anantapur, pp.57-61, 2005.
- 124. K. Swapna Sudha, A. Chennakesava Reddy, Tensile performance of heat treated AA2024/Al2O3 metal matrix composites using RVE models: experimental validation, National Conference on Advances in Design Approaches and Production Technologies (ADAPT-2005), Hyderabad, pp.332-334, 2005.
- 125. P. Arivazhagan, Moses Raja Cecil, A. Chennakesava Reddy, A. Rajadurai, Surface characteristics behaviour of the EDM eroded surface of Al-SiC metal matrix composite, National Conference on Computer Applications in mechanical Engineering, Anantapur, pp.165-171, 2005.
- 126. A. Chennakesava Reddy, Thermal Expansion Behavior of Aluminum Matrix Composites Reinforced with Fused Quartz Nanoparticles, National Conference on Advanced Materials and Manufacturing Techniques, Hyderabad, pp.350-355, 2004.
- 127. A. Chennakesava Reddy, Wear Characteristics of AA5050/TiC Metal Matrix Composites, National Conference on Advanced Materials and Manufacturing Techniques, Hyderabad, pp.356-360, 2004.
- 128. A. Chennakesava Reddy, Experimental evaluation of elastic lattice strains in the discontinuously SiC reinforced Al-alloy composites, National Conference on Emerging Trends in Mechanical Engineering, Nagapur, 2004.
- 129. S. Madhava Reddy, A. Chennakesava Reddy, Studies on machining characteristics of silicon nitride ceramics, National Conference on Advanced Materials and Manufacturing Techniques, Hyderabad, pp.88-90, 2004.
- 130. A. Chennakesava Reddy, Analysis of the Relationship Between the Interface Structure and the Strength of Carbon-Aluminum Composites, NATCON-ME, Bangalore, pp.61-62, 2004.
- 131. A. Chennakesava Reddy, Thermal Expansion Studies on Aluminum Matrix Composites with Different Reinforcement Volume Fractions of Si3N4 Nanoparticles, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.221-225, 2003.
- 132. A. Chennakesava Reddy, On the Wear of AA4015 Fused Silica Metal Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.226-230, 2003.
- 133. A. Chennakesava Reddy, B. Kotiveerachari, Effect of matrix microstucture and reinforcement fracture on the properties of tempered SiC/Al-alloy composites, National conference on advances in materials and their processing, Bagalkot, pp.121-124, 2003.
- 134. S. Madhav Reddy, A. Chennakesava Reddy, Clustering in Zirconium Oxide/AA1100 Alloy Particle-Reinforced Metal Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.182-187, 2003.
- 135. S. Madhav Reddy, A. Chennakesava Reddy, Effect of Particle Clustering on Micromechanical Properties of Boron Nitride/AA3003 Alloy Particle-Reinforced Metal

Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.188-192, 2003.

- 136. P. Laxminarayana, A. Chennakesava Reddy, Numerical Investigation of the Effect of Particle Clustering on the Micromechanical Properties of Titanium Nitride/AA4015 Alloy Particle-Reinforced Metal Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.193-196, 2003.
- 137. P. Laxminarayana, A. Chennakesava Reddy, Effect of Particle Spatial Distribution and Clustering on Tensile Behavior of Titanium Oxide/AA5050 Alloy Particle Reinforced Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.197-201, 2003.
- 138. Essa Zitoun, A. Chennakesava Reddy, Agglomeration of Nanoparticles into Network Aggregates in Zirconium Carbide/AA6061 Alloy Particle Reinforced Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.202-205, 2003.
- 139. A. Chennakesava Reddy, Finite Element Analysis Study of Micromechanical Clustering Characteristics of Graphite/AA7020 Alloy Particle Reinforced Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.206-210, 2003.
- 140. Essa Zitoun, A. Chennakesava Reddy, Unit Cell Models for Clustering of Particles embedded in MgO Particle/AA8090 Alloy Metal Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.211-215, 2003.
- 141. A. Chennakesava Reddy, Investigation of the Clustering Behavior of Titanium Diboride Particles in TiB2/AA2024 Alloy Metal Matrix Composites, 4th International Conference on Composite Materials and Characterization, Hyderabad, pp.216-220, 2003.
- 142. A. Chennakesava Reddy, Fracture behavior of brittle matrix and alumina trihydrate particulate composites, Indian Journal of Engineering & Materials Sciences, Vol.9, No.5, pp.365-368, 2002.
- 143. A. Chennakesava Reddy, Evaluation of Thermal Expansion of Al/B4C Metal Matrix Composites, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.196-200, 2002.
- 144. B. Kotiveera Chari, A. Chennakesava Reddy, Finite Element Modeling and Experimental Validation of Interphase Debonding and Particle Fracture in Titanium Carbide/AA1100 Alloy, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.156-161, 2002.
- 145. B. Kotiveera Chari, A. Chennakesava Reddy, Interphase Cracking in Titanium Nitride/2024 Alloy Particle-Reinforced Metal-Matrix Composites, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.162-167, 2002.
- 146. V. V. Satyanarayana, A. Chennakesava Reddy, Computation of Interphase Separation and Particle Fracture of Titanium Oxide/3003 Particle Reinforced Composites: The Role of Thermo-Mechanical Loading, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.168-173, 2002.
- 147. V. V. Satyanarayana, A. Chennakesava Reddy, Micromechanical Modeling of Reinforcement Fracture in Zirconium Carbide/4015 Particle-Reinforced Metal-Matrix

Composites, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.174-178, 2002.

- 148. A. Chennakesava Reddy, wo dimensional (2D) RVE-Based Modeling of Interphase Separation and Particle Fracture in Graphite/5050 Particle Reinforced Composites, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.179-183, 2002.
- 149. A. Chennakesava Reddy, Simulation of MgO/AA6061 Particulate-Reinforced Composites Taking Account of CTE Mismatch Effects and Interphase Separation, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.184-187, 2002.
- 150. Ch. Rajanna, A. Chennakesava Reddy, Effects of Interphase and Interface Characteristics on the Tensile Behavior of Boron Nitride/7020 Particle Reinforced Composites Subjected to Thermo-Mechanical Loading, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.188-191, 2002.
- 151. Ch. Rajanna, A. Chennakesava Reddy, Modeling of Interphases in SiO2/AA8090 alloy Particle -Reinforced Composites under Thermo-Mechanical Loading Using Finite Element Method, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.192-195, 2002.
- 152. A. Chennakesava Reddy, Two dimensional (2D) RVE-Based Modeling of Interphase Separation and Particle Fracture in Graphite/5050 Particle Reinforced Composites, 3rd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.179-183, 2002.
- 153. A. Chennakesava Reddy, Mechanisms of Load Transfer in Tension to Estimate Interfacial Behaviour of Kevlar 29 / Epoxy Composites by Laser Raman Spectroscopy, National Conference on Advances in Manufacturing Technologies (AMT-2001), Pune, pp.205-207, 2001.
- 154. H. B. Niranjan, A. Chennakesava Reddy, Computational Modeling of Interfacial Debonding in Fused Silica/AA7020 Alloy Particle-Reinforced Metal Matrix Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.222-227, 2001.
- 155. A. Chennakesava Reddy, Prediction of CTE of Al/TiB2 Metal Matrix Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.270-275, 2001.
- 156. H. B. Niranjan, A. Chennakesava Reddy, Nanoscale Characterization of Interfacial Debonding and Matrix Damage in Titanium Carbide/AA8090 Alloy Particle-Reinforced Metal Matrix Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.228-233, 2001.
- 157. A. Chennakesava Reddy, Significance of Testing Parameters on the Wear Behavior of AA1100/B4C Metal Matrix Composites based on the Taguchi Method, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.276-280, 2001.
- 158. S. Sundara Rajan, A. Chennakesava Reddy, Assessment of Temperature Induced Fracture in Boron Nitride/AA1100 Alloy Particle-Reinforced Metal Matrix Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.234-239, 2001.

JNT University, Hyderabad

- 159. S. Sundara Rajan, A. Chennakesava Reddy, Estimation of Fracture in Zirconia/AA2024 Alloy Particle-Reinforced Composites Subjected to Thermo-Mechanical Loading, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.240-245, 2001.
- 160. P. M. Jebaraj, A. Chennakesava Reddy, Finite Element Predictions for the Thermoelastic Properties and Interphase Fracture of Titanium Nitride /AA3003 Alloy Particle-Reinforced Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.246-251, 2001.
- 161. P. M. Jebaraj, A. Chennakesava Reddy, Effect of Thermo-Mechanical Loading on Interphase and Particle Fractures of Titanium Oxide /AA4015 Alloy Particle-Reinforced Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.252-256, 2001.
- 162. A. Chennakesava Reddy, Effect of CTE and Stiffness Mismatches on Interphase and Particle Fractures of Zirconium Carbide /AA5050 Alloy Particle-Reinforced Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.257-262, 2001.
- 163. A. Chennakesava Reddy, Behavioral Characteristics of Graphite /AA6061 Alloy Particle-Reinforced Metal Matrix Composites, 3rd International Conference on Composite Materials and Characterization, Chennai, pp.263-269, 2001.
- 164. Ch. Rajana, A. Chennakesava Reddy, Interfacial Reaction between Zirconium Alloy and Zirconia Ceramic Shell Mold, National Conference on Advanced Materials and Manufacturing Technologies, Hyderabad, pp.212-217, 2000.
- 165. S. Madhav Reddy, A. Chennakesava Reddy, Interfacial Reaction between Magnesium Alloy and magnesia Ceramic Shell Mold, National Conference on Advanced Materials and Manufacturing Technologies, Hyderabad, pp.218-222, 2000.
- 166. A. Chennakesava Reddy, Micromechanical and fracture behaviors of Ellipsoidal Graphite Reinforced AA2024 Alloy Matrix Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.96-103, 2000.
- 167. B. Kotiveera Chari, A. Chennakesava Reddy, Debonding Microprocess and interfacial strength in ZrC Nanoparticle-Filled AA1100 Alloy Matrix Composites using RVE approach, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.104-109, 2000.
- 168. S. Sundara Rajan, A. Chennakesava Reddy, Micromechanical Modeling of Interfacial Debonding in Silicon Dioxide/AA3003 Alloy Particle-Reinforced Metal Matrix Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.110-115, 2000.
- 169. S. Sundara Rajan, A. Chennakesava Reddy, Role of Volume Fraction of Reinforcement on Interfacial Debonding and Matrix Fracture in Titanium Carbide/AA4015 Alloy Particle-Reinforced Metal Matrix Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.116-120, 2000.
- 170. A. Chennakesava Reddy, Constitutive Behavior of AA5050/MgO Metal Matrix Composites with Interface Debonding: The Finite Element Method for Uniaxial Tension, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.121-127, 2000.

- 171. B. Kotiveera Chari, A. Chennakesava Reddy, Interfacial Debonding of Boron Nitride Nanoparticle Reinforced 6061 Aluminum Alloy Matrix Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.128-133, 2000.
- 172. P. M. Jebaraj, A. Chennakesava Reddy, Simulation and Microstructural Characterization of Zirconia/AA7020 Alloy Particle-Reinforced Metal Matrix Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.134-140, 2000.
- 173. P. M. Jebaraj, A. Chennakesava Reddy, Continuum Micromechanical modeling for Interfacial Debonding of TiN/AA8090 Alloy Particulate Composites, 2nd National Conference on Materials and Manufacturing Processes, Hyderabad, pp.141-145, 2000.
- 174. A. Chennakesava Reddy, Cohesive Zone Finite Element Analysis to Envisage Interface Debonding in AA7020/Titanium Oxide Nanoparticulate Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur-204, 209, 1999.
- 175. B. Kotiveera Chari, A. Chennakesava Reddy, Interfacial Debonding Analysis in Nanoparticulate Reinforced Metal Matrix Composites of AA8090/Zirconium Carbide, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.210-214, 1999.
- 176. B. Kotiveera Chari, A. Chennakesava Reddy, Effect of Debonding on Overall Behavior of AA3003/Titanium Carbide Nanoparticulate Reinforced Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.220-224, 1999.
- 177. P. M. Jebaraj, A. Chennakesava Reddy, Analysis of Debonding along Interface of AA4015/Magnesium Oxide Nanoparticulate Reinforced Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.225-229, 1999.
- 178. H. B. Niranjan, A. Chennakesava Reddy, Effect of Particulate Debonding in AA5050/Boron Nitride Nanoparticulate Reinforced Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.230-234, 1999.
- 179. P. M. Jebaraj, A. Chennakesava Reddy, Interface Debonding Prediction Technique for Tensile Loaded AA6061/Zirconium Oxide Nanoparticulate MMC, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.235-239, 1999.
- 180. S. Sundara Rajan, A. Chennakesava Reddy, FEM Model for Volume Fraction Dependent Interface Debonding in TiN Nanoparticle Reinforced AA7020 Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.240-244, 1999.
- 181. A. Chennakesava Reddy, Micromechanical Modelling of Interfacial Debonding in AA1100/Graphite Nanoparticulate Reinforced Metal Matrix Composites, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.249-253, 1999.
- 182. B. Kotiveerachari, A. Chennakesava Reddy, Interfacial effect on the fracture mechanism in GFRP composites, CEMILAC Conference, Ministry of Defence, India, Vol No.1, Issue No.B, pp.85-87, 1999.

- 183. S. Sundara Rajan, A. Chennakesava Reddy, Deformation Behavior of AA8090/ TiO2 Nanoparticulate Reinforced Metal Matrix Composites with Debonding Interfaces, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.245-248, 1999.
- 184. H. B. Niranjan, A. Chennakesava Reddy, Debonding Failure and Volume Fraction Effects in Nano-reinforced Composites of AA2024/Silicon Oxide, 2nd International Conference on Composite Materials and Characterization, Nagpur, pp.215-219, 1999.
- 185. A. Chennakesava Reddy, Assessment of Debonding and Particulate Fracture Occurrences in Circular Silicon Nitride Particulate/AA5050 Alloy Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.104-109, 1998.
- 186. H. B. Niranjan, A. Chennakesava Reddy, Effect of Elastic Moduli Mismatch on Particulate Fracture in AA7020/Silicon Nitride Particulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.115-118, 1998.
- 187. P. M. Jebaraj, A. Chennakesava Reddy, Cohesive Zone Modelling for Interface Debonding in AA8090/Silicon Nitride Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.119-122, 1998.
- 188. P. M. Jebaraj, A. Chennakesava Reddy, Plane Strain Finite Element Modeling for Interface Debonding in AA1100/Silicon Oxide Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.123-126, 1998.
- 189. A. Chennakesava Reddy, Local Stress Differential for Particulate Fracture in AA2024/Titanium Carbide Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.127-131, 1998.
- 190. B. Kotiveera Chari, A. Chennakesava Reddy, Interface Debonding and Particulate Facture based on Strain Energy Density in AA3003/MgO Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.132-136, 1998.
- 191. H. B. Niranjan, A. Chennakesava Reddy, Numerical and Analytical Prediction of Interface Debonding in AA4015/Boron Nitride Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.137-140, 1998.
- 192. S. Sundara Rajan, A. Chennakesava Reddy, Effect of Particulate Volume Fraction on Particulate Cracking in AA5050/Zirconium Oxide Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.156-159, 1998.
- 193. S. Sundara Rajan, A. Chennakesava Reddy, Cohesive Zone Analysis for Interface Debonding in AA6061/Titanium Nitride Nanoparticulate Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.160-164, 1998.
- 194. B. Kotiveera Chari, A. Chennakesava Reddy, Numerical Simulation of Particulate Fracture in Round Silicon Nitride Particulate/AA6061 Alloy Metal Matrix Composites, National Conference on Materials and Manufacturing Processes, Hyderabad, pp.110-114, 1998.

- 195. S. Sundara Rajan, A. Chennakesava Reddy, Evaluation of Tensile Behavior of Boron Carbide/AA1100 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.156-159, 1997.
- 196. S. Sundara Rajan, A. Chennakesava Reddy, Assessment of Tensile Behavior of Boron Carbide/AA2024 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.160-163, 1997.
- 197. P. Martin Jebaraj, A. Chennakesava Reddy, Prediction of Tensile Behavior of Boron Carbide/AA3003 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.164-166, 1997.
- 198. A. Chennakesava Reddy, Effect of Particle Loading on Microealstic Behavior and interfacial Tractions of Boron Carbide/AA4015 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.176-179, 1997.
- 199. B. Kotiveera Chari, A. Chennakesava Reddy, Estimation of Micro-stresses and interfacial Tractions in Boron Carbide/AA5050 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.180-182, 1997.
- 200. P. Martin Jebaraj, A. Chennakesava Reddy, Prediction of Micro-stresses and interfacial Tractions in Boron Carbide/AA6061 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.183-185, 1997.
- 201. B. Kotiveera Chari, A. Chennakesava Reddy, Computation of Micro-stresses and interfacial Tractions in Boron Carbide/AA7020 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.186-188, 1997.
- 202. H. B. Niranjan, A. Chennakesava Reddy, Valuation of Micro-stresses and interfacial Tractions in Boron Carbide/AA8090 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.189-191, 1997.
- 203. H. B. Niranjan, A. Chennakesava Reddy, Determination of Micro-stresses and interfacial Tractions in Titanium Boride/AA1100 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.192-194, 1997.
- 204. A. Chennakesava Reddy, Reckoning of Micro-stresses and interfacial Tractions in Titanium Boride/AA2024 Alloy Metal Matrix Composites, 1st International Conference on Composite Materials and Characterization, Bangalore, pp.195-197, 1997.
- 205. A. Chennakesava Reddy, Interfacial Debonding Analysis in Terms of Interfacial Tractions for Titanium Boride/AA3003 Alloy Metal Matrix Composites, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.124-127, 1997.
- 206. P. Martin Jebaraj, A. Chennakesava Reddy, Effect of Interfacial Debonding on Stiffness of Titanium Boride/AA5050 Alloy Metal Matrix Composites, National Conference on Modern Materials and Manufacturing, Pune, pp.132-135, 1997.
- 207. S. Sundara Rajan, A. Chennakesava Reddy, Micromechanical modeling of Titanium Boride/AA7020 Alloy Metal Matrix Composites in Finite Element Analysis using RVE Model, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.140-143, 1997.

- 208. P. Martin Jebaraj, A. Chennakesava Reddy, Effect of Interfacial Tractions of Rectangular Titanium Boride Particulate/AA8090 Alloy Metal Matrix Composites, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.144-147, 1997.
- 209. S. Sundara Rajan, A. Chennakesava Reddy, Cohesive Zone interfacial debonding of Silicon Nitride/AA1100 Alloy Metal Matrix Composites Using Finite Element Analysis, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.265-268, 1997.
- 210. S. Sundara Rajan, A. Chennakesava Reddy, Simulation of Micromechanics for interfacial debonding in Silicon Nitride/AA2024 Alloy Metal Matrix Composites, 1st National Conference on Modern Materials and Manufacturing, Pune-269, 272, 1997.
- 211. P. Martin Jebaraj, A. Chennakesava Reddy, Finite Element Analysis for Assessment of Dislocation and Debonding Events in Silicon Nitride/AA3003 Alloy Metal Matrix Composites, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.273-277, 1997.
- 212. A. Chennakesava Reddy, Evaluation of Debonding and Dislocation Occurrences in Rhombus Silicon Nitride Particulate/AA4015 Alloy Metal Matrix Composites, 1st National Conference on Modern Materials and Manufacturing, Pune, pp.278-282, 1997.
- 213. Alaneme, K. K. and Aluko, A. O., (2012), Fracture toughness (K1C) and tensile properties of as-cast and age-hardened aluminium (6063)-silicon carbide particulate composites, Scientia Iranica, vol. 19(4), pp. 992–996.
- 214. Anthony, M., Schultz, B.F. and Pradeep Rohatgi, (2012), Metal matrix composites offer the automotive industry an opportunity to reduce vehicle weight, improve performance, Advanced Materials, and Processes, vol. 3, pp. 19-23.
- 215. Bakunin, V. N., Suslov, A. Y. Y., Kuzmina, G. N., Parenago, O. P. and Topchiev, V., (2004), Synthesis and application of inorganic nanoparticulates as lubricant components a Review, Journal of Nanoparticulate Research, vol. 6, pp. 273–284.
- 216. Bayraktar, E. and Katundi, D., (2010), Development of a new aluminium matrix composite reinforced with iron oxide (Fe3 O4), Journal of Achievements in Materials and Manufacturing Engineering, vol. 38(1), pp. 7-14.
- 217. Boopathi, M. M., Arulshri, K.P. and Iyandurai, N., (2013), Evaluation of mechanical properties of aluminium alloy 2024 reinforced with silicon carbide and flyash hybrid metal matrix composite, American Journal of Applied Sciences, vol. 10(3), pp. 219–229.
- 218. Chawla, N. and Shen, Y. L., (2001), Mechanical behaviour of particle reinforced metal matrix composites, Advanced Engineering Materials, vol. 3(6), pp. 357-370.
- 219. Cocen, U. and Onel, K., (2002), Ductility and strength of extruded SiCp/aluminiumalloy composites, Composites Science and Technology, vol. 62, pp. 275–282.
- 220. Davis, J.R., (1993), Aluminum Matrix Composites, ASM speciality handbook: Aluminum & Aluminum Alloy, ASM International, pp. 160-179.
- 221. Foltz, J. V., and Blackmon, C. M., (1990), Metal matrix composites, In ASM

International Metals Handbook, Properties and Selection: Nonferrous Alloys and Special-

Purpose Materials, vol.10(2), pp. 903-912.

- 222. Foltz, J.V. and Blackmon, C.M., (2001), Metal matrix composites, ASM Handbook: Composites, vol. 1(21), pp. 903-912.
- 223. Greenwood, N. N. and Earnshaw, A., (1997), Chemistry of the Elements, Butterworth-

Heinemann, ISBN: 978-0-08-037941-8, p.297.

- 224. Hashim, J., Looney, L. and Hashmi, M.S.J., (2002), Particle distribution in cast metal matrix composites-Part I, Journal of Material Processing Technology, vol. 123, pp. 251-257.
- 225. Jing, J. K., Ji, X. L. and Jiang, B. Z., (2002), Tensile modulus of polymer nanocomposites, Polymer Engineering, and science, vol. 42, pp. 983-993.

- 226. Kolsgaard, A., Arnberg, L. and Brusethaug, S., (1993), Solidification microstructures of AlSi7Mg-SiC particulate composite, Materials Science and Engineering: A, vol.173 (1–2), pp. 243-250.
- 227. Mazaherin, Y., Martian, M., Emadi, R. and Najarian, A.R., 2013, Comparison of microstructural and mechanical properties of Al–TiC, Al–B4C and Al–TiC-B4C composites prepared by casting techniques, Materials Science & Engineering A, vol. 560, pp. 278–287.
- 228. Surappa, M.K., and Rohatgi, P.K., (1981), Preparation and properties of aluminium alloy ceramic particle composites, Journal of Materials Science, vol. 16, pp. 983–993.
- 229. Tevet, O., Palle, V. H., Ronit, P. B., Rita R., Daniel W. H, and Tenne, R., (2011), Friction mechanism of individual multilayered nanoparticulates, Proceedings of the National Academy of Sciences of the United States of America, doi:10.1073/pnas.1106553108.
- 230. Zhu, S. J., Peng, L. M., Zhou, Q., Ma, Z.Y., Kucharova, K.and Cadek, J., (1998),

Creep behaviour of aluminium strengthened by fine aluminium carbide particles and reinforced by silicon carbide particulates DS Al-SiC/Al4C3composites, Acta Technica,

vol.5, pp.435-455.

231. Abhishek K., Shyam Lal, and Sudhir Kumar, (2013), Fabrication and characterization of A359/Al2O3 metal matrix composite using electromagnetic stir casting method,

Journal of Materials Research and Technology, vol. 2(3), pp. 250-254.

- Aggour, L., Fitzer, E., Heym, M. and Ignatowitz, E., (1977), Thin coatings on carbon fibers as diffusion barriers and wetting agents in Al composites, Thin Solid Films, vol. 40, pp.97-105.
- 233. Alblas J. B., and Kuipers M., (1970), The contact problem of a rigid cylinder rolling on

a thin viscoelastic layer, International Journal of Engineering Science, 8(5), pp. 363-

380.

- 234. Arsenault, R., (1984), The strengthening of aluminium alloy 6061 by fiber and platelet silicon carbide, Material Science Engineering, vol. 64(2), pp. 171-181.
- 235. Asthana, R., (1998), Reinforced cast metals: part-I Solidification microstructure, Journal of Materials Science, vol. 33(7), pp. 1679–1698.
- 236. Baradeswaran, A. and Perumal, E.,(2013), Influence of B4C on the tribological and mechanical properties of Al 7075–B4C composites, Composites: Part B, vol. 54, pp. 146–152.
- 237. Basavarajappa, S. and Chandra Mohan, G., (2005), Wear studies on metal matrix composites a Taguchi approach, Journal of material science technology, vol. 21, pp. 845-890.
- 238. Bruzzi, M.S., McHugh, P.E., Rouke, F, and Linder, T.,(2001), Micromechanical modelling of the static and cyclic loading of an Al 2124-SiC MMC, International Journal of plasticity, vol.17, pp. 565-599.
- 239. Caleb, C.G., Ivanovich, E.G., Claudia, L.M., Ernesto, L.S., Rubén C.B., Raul P.B., and Jose Martin H.R., (2018), B4C particles reinforced Al2024 composites via mechanical milling, Metals, vol.8, p. 647
- 240. Campbell, F.,(2006), Manufacturing technology for aerospace structural materials,1st

ed., Elsevier Science, Amsterdam, Netherlands.

- 241. Cerit, A.A., Karamis, M.B., Nair, F. and Yildizl, K., (2008), Effect of reinforcement particle size and volume fraction on wear behaviour of metal matrix composites, Tribology in Industry, vol. 30, pp. 31-36.
- 242. Chawla, N. and Shen, Y. L., (2001), Mechanical behaviour of particle reinforced metal matrix composites, Advanced Engineering Materials, vol. 3(6), pp. 357-370.
- 243. Chen, L. H., and Rigney, D. A., (1985), Transfer during unlubricated sliding wear of selected metal systems, Wear. vol. 105(1), pp. 47–61.
- 244. Chernyshova and Rebrov, A., (1986), Interaction kinetics of boron carbide and silicon carbide with liquid aluminium, Journal of the Less Common Metals, vol. 117(1-2), pp. 203-207.
- 245. Foltz, J. V., and Blackmon, C. M., (1990), Metal matrix composites, In ASM International, Metals Handbook, Properties, and Selection: Nonferrous Alloys and Special-Purpose Materials, vol. 10 (2), pp. 903-912.
- 246. Ghosh, S. and Saha, P., (2011), Crack and wear behaviour of SiC particulate reinforced aluminium-based metal matrix composite fabricated by direct metal laser sintering process, Materials and Design, vol. 32, pp. 139-145.
- 247. Ghosh, S., Novack, Z. and Lee, K., (1997), Quantitative characterization and modelling of composite microstructures by Voronoi cells, Acta Materialia, vol. 45(6), pp. 2215-2234.
- 248. Gopinath, (1979), Proceedings of National conference on aluminium metallurgy, IIsc Bangalore, India, pp.289-294.
- 249. Harnby, N., Edward, M. F. and Nienow, A. W., (1985), Mixing in Process Industries, Butterworths, London.
- 250. Harrigan, W.C.,(1998), Commercial processing of metal matrix composites, Materials Science and Engineering: A, vol. 244, pp. 75–79.
- 251. Hashim, J., Looney, L., and Hashmi, M. S. J., (1999), Metal matrix composites: production by the stir casting method, Journal of Materials Processing Technology, vol. 92–93, pp. 1–7.
- 252. Hashim, J., Looney, L., and Hashmi, M.S.J.,(2001), The enhancement of wettability of SiC particles in cast aluminium matrix composites, Journal of Materials Processing Technology, vol.119(1-3), pp. 329-335.
- 253. Hashim, J.,(2012), The production of cast metal matrix composite by a modified stir casting method, Journal Technology, vol. 35(1), pp. 9-20.
- 254. Hecht, U. and Rex, S., (1997), On the transition from pushing to engulfment during directional solidification of the particle-reinforced aluminium-based metal-matrix

composite 2014+11%vol Al2O3, Metallurgical and Materials Transactions A, vol. 28(13), pp. 867–874.

- 255. Hong, C.P., Shen, H.F. and Cho, I.S.,(1998), Prevention of macrosegregation in the squeeze casting of an Al-4.5%wt Cu alloy, Metallurgical and Materials Transactions A, vol. 29(1), pp. 339–349.
- 256. Hosking, F.M., Portillo, F.F., Wunderlin, R. and Mehrabian, R., (1982), Composite of Al alloys: fabrication and wear behaviour, J. Mater. Sci., vol. 17, pp. 477–98.
- 257. Iseki, T., Kameda, T., and Maruyama, T., (1984), Interfacial reactions between SiC and aluminium during joining, Journal of Material Science, vol. 19(5), pp. 1692-1698.
- 258. Ishikawa, T., Tanaka, J., Teranishi, H., Okamura, T. and Hayase, T., (1984), Process for the surface treatment of inorganic fibers for reinforcing titanium or nickel and product, U.S. Patent No. 4, 440, 571.
- 259. Jakab, E. and Omastova, M., (2005), Thermal decomposition of polyolefin/carbon black composites, Journal of Analytical and Applied Pyrolysis, vol. 74(1-2), pp. 204-214.
- 260. Kainer, K., (2006), Metal Matrix Composites, Custom-made materials for automotive

and Aerospace Engineering, John Wiley & Sons, Inc. Hoboken, NJ.

- 261. Kalaiselvan, K., Murugan, N. and Parameswaran, S.,(2011), Production and characterization of AA6061–B4C stir cast composite, Materials, and Design, vol. 32, pp. 4004–4009.
- 262. Kennedy, A.R., Weston, D.P. and Jones, M.I., (2001), Reaction in Al–TiC metal matrix composites, Materials Science and Engineering: A, vol. 316(1–2), pp. 32-38.
- 263. Khalil, H.A., Firoozian, P., Bakare, I.O., Akil, H.M. and Noor, A.M., (2010), Exploring biomass-based carbon black as filler in epoxy composites: Flexural and thermal properties. Materials & Design, vol. 31(7), pp. 3419-3425.
- 264. Leggoe, J.W; Mammoli, A.A; Bush, M.B; Hu, X.Z., (1998), Finite element modelling of deformation in particulate reinforced metal matrix composites with random local microstructure variation; Acta mater. vol. 46(17), pp. 6075-6088.
- Li, J.L. and Xiong, D.S., (2008), Tribological properties of nickel-based self-lubricating composite at elevated temperature and counter face material selection. Wear, vol. 265(3), pp. 533- 539.
- 266. Liu, Y., Rohatgi, P.K. and Ray, S., (1993), Tribological characteristics of aluminum-50& Vol. graphite composite. Metallurgical Transactions A, vol. 24(1), pp. 151-159.
- 267. Luo, A., (1995), Microstructure and mechanical behaviour of cast magnesium metal matrix composites, Metallurgical and Materials Transactions A, vol. 26, pp. 2445–2455.
- 268. Macke, A., Schultz, B. and Rohatgi, P., (2012), Metal Matrix Composites, Advanced Materials & Processes, 170 (3), pp 19 23.
- 269. Michael, R.W., (2007), Factors affecting the transverse tensile strength of unidirectional continuous silicon carbide fiber-reinforced 6061 Aluminium, Journal of composite materials. pp. 707-726.

- 270. Mimoto, T., Nakanishi, N., Umeda, J. and Kondoh, K., (2011), Mechanical properties and strengthening mechanism of pure Ti powder composite material reinforced with carbon nanoparticulates, Transactions of Joining and Welding Research Institute (JWRI), vol. 40(2), pp. 63-68.
- 271. Mori, T. and Tanaka, K., (1973), Average stress in matrix and average elastic energy of materials with misfitting inclusions, Acta Metallurgica, vol. 21(5), pp. 571-574.
- 272. Nogi, K., (2010), The role of wettability in metal-ceramic joining, Scripta Materialia, vol. vol. 62, pp. 945-948.
- 273. Ozdemir, I., Toparli, M., Onel, K. and Tsunekawa, Y., (2005), Fracture and Failure of

Al-SiCp composites at different temperatures and conditions, Journal of Composite

materials, vol. 39, pp. 601-615.

- 274. Patnaik, S., Swain, P., Mallik, P. and Sahoo, S.,(2014), Wear characteristics of aluminium-graphite composites produced by stir casting technique. Journal of Metallurgical and Materials Engineering, Vol.4, pp.13–20.
- 275. Ramu, G. and Ranjit, B., (2009), Effect of equal channel angular pressing (ECAP) on

microstructure and properties of Al-SiCp composites, Materials & Design, vol. 30(9), pp.

3554-3559.

- 276. Rohatgi, P.K., Ray, S. and Liu, Y., (1992), Tribological properties of metal matrixgraphite particle composites, International Materials Reviews, vol. 37(3), pp. 129–149.
- 277. Rohatgi, P.K., Menezes, P.L. and Lovell, M.R.,(2012), Self-lubricating behaviour of graphite-reinforced metal matrix composites, Green tribology, Springer, New York, pp 445–480.
- 278. Rohatgi, P.K., Afsaneh, D.M., Schultz, B.F. and Ferguson, J., (2013), Synthesis and properties of metal matrix nanocomposites (MMNCS), syntactic foams, self-lubricating, and self-healing metals. In: PRICM: 8 Pacific Rim international congress on advanced materials and processing. Wiley, New York, 1515–1524.
- 279. Sajjadi, S.A., Ezatpour, H.R. and Beygi, H.,(2011), Microstructure and mechanical properties of Al-Al2O3 micro and nanocomposites fabricated by stir casting, Materials Science and Engineering A, vol. 528, pp. 8765–8771.
- 280. Sambath Kumar, M., Navaneetha Krishnan, P., Ponappa, K. and Sasikumar, K.S.K.,

(2017), Mechanical and corrosion behaviour of Al7075 (Hybrid) metal matrix composites

by two-step stir casting process, Latin American Journal of Solids and Structures, http://dx.doi.org/10.1590/1679-78253132.

- 281. Shen, H. and Missenden, C.J., (2002), 3D finite element analysis of particle reinforced aluminium; Material Science and Engineering A, vol. 338, pp. 271-281.
- 282. Surappa, M. K., (2003), Aluminium metal matrix composites: challenges and

opportunities, Sadhana, vol. 28(1-2), pp. 319-334.

- 283. Tiwari, S.N., Pathak, J.P. and Malhotra, S.L., (1979), Production of high leaded Al by impeller mixing, Metals Technology, vol. 6, pp. 442-445.
- 284. Veeresh Kumar, G.B., Rao, C.S.P., Selvaraj, N., and Bhagyashekar, M.S., (2009), Studies on AL 6061-Sic and Al 7075-Al2o3 metal matrix composites, Journal of minerals and materials characterization and Engineering, vol. 9, PP 43-55.

285. Warren, R. and Anderson, C., (1984), Silicon carbide fibers and their potential for use in composite materials, Part II, Composites, vol. 15(2), pp. 101-111.

286. Wu, L.L., Yang, W.J., Xu, J.R. and Yao, G.C.,(2013), Wear resistance of graphite/aluminium composites that prepared by stirring casting, Advanced Materials Research, vol. 638, pp. 333–338.

287. Yue, J., Xu, Y. and Bao, J., (2017), Epoxy–carbon black composite foams with tunable

electrical conductivity and mechanical properties: Foaming improves the conductivity,

Journal of Applied Polymer Science, vol. 134 (33), pp. 45071.

- 288. Zhang, D., Sugio, K. and Sakai, K., Fukushima, H. and Yanagisawa, O., (2007), Effect of the spatial distribution of SiC particles on the tensile deformation behaviour of Al-10vol%SiC composites, Materials Transactions, vol. 48(2), pp. 171 to 177.
- Zhang Y., Chromik R.R., (2018), Tribology of self-lubricating metal matrix composites. In: Menezes P., Rohatgi P., Omrani E. (eds.) Self-Lubricating Composites. Springer, Berlin, Heidelberg, pp 33-73.